Author's response to reviews

Title: Evaluation of a web-based intervention to reduce antibiotic prescribing for LRTI in six European countries; quantitative process analysis of the GRACE/INTRO randomised controlled trial.

Authors:

Lucy Yardley (L.Yardley@soton.ac.uk)
Elaine Douglas (Elaine.Douglas.11@ucl.ac.uk)
Sibyl Anthierens (Sibyl.Anthierens@ua.ac.be)
Sarah Tonkin-Crine (S.K.Tonkin-Crine@soton.ac.uk)
Gilly O'Reilly (G.O'Reilly@soton.ac.uk)
Beth Stuart (bhs1@soton.ac.uk)
Adam W Geraghty (A.W.Geraghty@soton.ac.uk)
Emily Arden-Close (E.J.Arden-Close@soton.ac.uk)
Alike W van der Velden (A.W.vanderVelden@umcutrecht.nl)
Herman Goosens (Herman.Goossens@uza.be)
Theo JM Verheij (t.j.m.verheij@umcutrecht.nl)
Chris C Butler (butlerCC@cardiff.ac.uk)
Nick A Francis (francisna@cardiff.ac.uk)
Paul Little (P.Little@soton.ac.uk)

Version: 3
Date: 7 November 2013

Author's response to reviews:

The changes I have rejected in the copy-edited version are:

1. 'on behalf of the GRACE/INTRO consortium' does need to appear at the end of the author list, as it refers to the whole author list and this is the conventional place to acknowledge the wider research group.

2. There is no numbered street address for University of Southampton - it is a campus.

3. Reporting on p.14 was corrected back to reported, which is accurate and grammatically correct.

4. The copy-editor queried the reporting of degrees of freedom in a couple of places, but this was due to misunderstanding what they were - the original reporting format was correct.